# UC Davis UC Davis Previously Published Works

## Title

Prenatal exposure to pesticide residues in the diet in association with child autismrelated traits: Results from the EARLI study

**Permalink** https://escholarship.org/uc/item/4f90b2dg

**Journal** Autism Research, 15(5)

**ISSN** 1939-3792

## **Authors**

Joyce, Emily E Chavarro, Jorge E Rando, Juliette <u>et al.</u>

Publication Date 2022-05-01

## DOI

10.1002/aur.2698

Peer reviewed

DOI: 10.1002/aur.2698

## RESEARCH ARTICLE

# Prenatal exposure to pesticide residues in the diet in association with child autism-related traits: Results from the EARLI study

Emily E. Joyce<sup>1</sup> || Jorge E. Chavarro<sup>2</sup> | Juliette Rando<sup>3</sup> | Ashley Y. Song<sup>4</sup> | Lisa A. Croen<sup>5</sup> || M. Daniele Fallin<sup>4</sup> || Irva Hertz-Picciotto<sup>6,7</sup> || Rebecca J. Schmidt<sup>6,7</sup> || Heather Volk<sup>4</sup> || Craig J. Newschaffer<sup>8</sup> || Kristen Lyall<sup>1,3</sup> ||

<sup>1</sup>Dornsife School of Public Health, Department of Epidemiology and Biostatistics, Drexel University, Philadelphia, Pennsylvania, USA

<sup>2</sup>Department of Nutrition, Harvard T.H. Chan School of Public Health, Boston, Massachusetts, USA

<sup>3</sup>A.J. Drexel Autism Institute, Drexel University, Philadelphia, Pennsylvania, USA

<sup>4</sup>Department of Mental Health, Johns Hopkins, Baltimore, Maryland, USA

<sup>5</sup>Division of Research, Kaiser Permanente Northern California, Oakland, California, USA

<sup>6</sup>Department of Public Health Sciences, University of California, Davis, California, USA

<sup>7</sup>MIND Institute, Sacramento, CA, USA

<sup>8</sup>College of Health and Human Development, Penn State, University Park, State College, Pennsylvania, USA

#### Correspondence

Kristen Lyall, Dornsife School of Public Health, Department of Epidemiology and Biostatistics, Drexel University, 3215 Market St, Philadelphia, PA 19104, USA. Email: kld98@drexel.edu

#### Funding information

Autism Speaks, Grant/Award Number: AS 5938; Drexel University Institutional Review Board, Grant/Award Numbers: 17862, 71109; Drexel University Department of Epidemiology; National Institute of Child Health and Human Development; National Institute of Mental Health; National Institute of Environmental Health Sciences, Grant/Award Number: R01 ES016443; P30DK046200; Dornsife School of Public Health Department of Epidemiology Pilot Funding Series

#### Abstract

Prior work has suggested associations between prenatal exposure to several classes of pesticides and child autism spectrum disorder (ASD). We examined a previously developed pesticide residue burden score (PRBS) and intake of high pesticide residue foods in association with ASD-related traits. Participants were drawn from the Early Autism Risk Longitudinal Investigation (EARLI) (n = 256), a cohort following mothers who previously had a child with ASD through a subsequent pregnancy and that child's development. ASD-related traits were captured according to total Social Responsiveness Scale (SRS) scores at age 3 (mean raw total SRS score = 35.8). Dietary intake was assessed through a food frequency questionnaire collected during pregnancy. We also incorporated organic intake and fatty foods in modified versions of the PRBS. Associations between highresidue fruit and vegetable intake, the overall PRBS and modified versions of it, and SRS scores were assessed using multivariable linear regression. Overall, we did not observe associations between pesticide residues in foods and ASD-related outcomes, and modified versions of the PRBS yielded similar findings. However, reductions in ASD-related traits were observed with higher overall fruit and vegetable intake (adjusted estimates for O4 vs. O1:  $\beta$  -12.76, 95%CI -27.8, 2.3). Thus, findings from this high familial probability cohort did not suggest relationships between pesticide residues in the diet according to the PRBS and ASDrelated traits. Beneficial effects of fruit and vegetable intake may influence these relationships. Future work should consider fruit and vegetable intake in association with ASD-related outcomes.

Lay Summary: Diet is the main source of exposure to most pesticides in use today. In this study, we examined the relationship between pesticide exposure from residues in the diet during pregnancy and child autism-related traits. We found that these pesticide residues from the diet were not related to child autism-related outcomes at age three. However, higher prenatal fruit and vegetable intake was associated with reductions in child autism-related traits.

#### **KEYWORDS**

autism spectrum disorder, autism-related traits, fruit, pesticide residues, prenatal diet, vegetables

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made. © 2022 The Authors. *Autism Research* published by International Society for Autism Research and Wiley Periodicals LLC.

## INTRODUCTION

Pesticides have the ability to cross the placenta (Bradman et al., 2003; Caspersen et al., 2016; Dewan et al., 2013; Wolff et al., 2005), as well as disrupt hormones involved in neurodevelopment (C. Li et al., 2014; Wang et al., 2017). Although certain classes of pesticides have been banned or limited in Western countries, including organochlorine (OC) pesticides, there remains a need to examine the impact of these chemicals due to their long half-lives of up to 10 years in lipids, and bioaccumulation in the environment and food chain (Fourth National Report on Human Exposure to Environmental Chemicals, 2009; Jayaraj et al., 2016; Y.-F. Li, n.d.). While newer replacement classes of pesticides, such as organophosphate (OP) pesticides and pyrethroids, do not have these lipophilic properties, their widespread use also makes them relevant to public health considerations. Recent biomonitoring studies demonstrate detectable levels of both newer and older classes of pesticides, both in the general population and in pregnant women (Fourth National Report on Human Exposure to Environmental Chemicals, 2009; Lyall, Croen, Sjödin, et al., 2017; Woodruff et al., 2011).

Evidence has suggested a relationship between prenatal pesticide exposure and cognitive and developmental deficits in children (Bouchard et al., 2010; Engel et al., 2016; Eskenazi et al., 2007; Furlong et al., 2017; Korrick & Sagiv, 2008; Rosas & Eskenazi, 2008), and emerging work has also suggested pesticide exposure may be associated with autism spectrum disorder (ASD) (Brown et al., 2018; Lyall, Croen, Sjödin, et al., 2017; Philippat et al., 2018; Sagiv et al., 2018). ASD is a complex neurodevelopmental condition characterized by social communication deficits and challenges manifesting at a young age (Diagnostic and statistical manual of mental disorders: DSM-V, 2013). There is an abundance of evidence that suggests ASD etiology derives from a combination of genetic and environmental factors, with the prenatal stage as a susceptibility period (Hallmaver et al., 2011; Risch et al., 2014).

Only a handful of studies to date have addressed the association between prenatal pesticide exposure and ASD-related outcomes specifically. Early studies examined residential proximity to agricultural exposure to pesticides and reported increased risk of ASD with exposure to certain OC (Roberts et al., 2007) and OP pesticides (Shelton et al., 2014). Two more recent studies examining close proximity to agricultural applications have shown no association with ASD (Sagiv et al., 2018; Schmidt et al., 2017), while the largest study to date using this exposure metric found associations with multiple individual non-persistent pesticides and close proximity to agriculture applications (von Ehrenstein et al., 2019). A few studies using measured prenatal levels to determine exposure have found evidence for associations between higher exposure to certain OC, OP, and pyrethroid pesticides

and increased risk for ASD and ASD-related traits (Braun et al., 2014; Brown et al., 2018; Furlong et al., 2017; Millenson et al., 2017; Sagiv et al., 2018; Schmidt et al., 2012), though null results have also been reported (Barkoski et al., 2021; Lyall, Croen, Sjödin, et al., 2017; Philippat et al., 2018). To date, limited work has considered combined effects of pesticides in association with ASD and/or child neurodevelopment, though pesticides are known to co-occur (Fourth National Report on Human Exposure to Environmental Chemicals, 2009; Hamra et al., 2019; Woodruff et al., 2011). In addition, work from animal studies and other fields suggests the potential for mixture effects that are not fully explained by analyses of individual pesticides alone, suggesting that existing work considering individual pesticides or classes could misrepresent actual exposures and associations (European Commission, 2012; Laetz et al., 2009; Moser et al., 2006). Yet, measurement across multiple classes is costly and may not align with sample-sparing requirements of studies with limited prenatal biological specimens, which are unique resources in studies of ASD.

Diet, particularly fruit and vegetable intake, is the primary source of exposure to major pesticide classes in the general population (Cequier et al., 2017; Fourth National Report on Human Exposure to Environmental Chemicals, 2009; Morgan, 2012). The US Pesticide Monitoring Program has reported that fruits and vegetables contain the highest proportion of detected pesticide levels across all foods annually tested (Pesticide Data Program Annual Summary, Calendar Year 2013, 2014). Organic foods also contain detected pesticide residues, although at a lesser amount compared to conventional produce (Mie et al., 2017). Recently, a questionnaire-based measure that captures approximately 450 pesticide residues in the diet has been developed and validated against urinary pesticide levels in two independent populations (Chiu, Williams, Mínguez-Alarcón, Gillman, et al., 2018; Hu et al., 2016). In this way, the measure is an efficient metric of combined exposure to pesticides. This pesticide residue burden score (PRBS) has been used in studies examining associations with pregnancy, fetal growth, and live birth outcomes (Chiu, Williams, Gillman, Gaskins, et al., 2018; Chiu, Williams, Gillman, Hauser, et al., 2018). Yet, no prior study has examined whether pesticide residues in the diet may be associated with ASD and related child outcomes. Given challenges with collecting biospecimens in key suspected critical time periods for ASD, cost considerations, and specimen volume requirements associated with pesticide measurement across multiple classes, a questionnaire-based exposure assessment represents an efficient and accessible way to capture pesticide exposure across many classes and address this relationship in ASD research.

In this study, we sought to examine the relationship between diet-based pesticide exposure and ASD-related outcomes, including quantitative ASD-related traits and cognitive scores, using prospectively collected dietary data from a high-familial probability cohort. We also sought to test modifications to the PRBS to account for other food sources of pesticides, as well as organic food status. Based on prior work suggesting links between prenatal pesticide exposure and ASD diagnosis and/or poorer neurodevelopmental outcomes, we hypothesized that higher prenatal pesticide exposure in the diet would be associated with increases in ASD-related traits and decreases in cognitive scores.

## **METHODS**

## Study population

Participants in this study were from the Early Autism Risk Longitudinal Investigation (EARLI) (Newschaffer et al., 2012). EARLI is a high-familial probability cohort study that enrolled women who already had a child with a confirmed ASD diagnosis and followed participants through a subsequent pregnancy and the first 3 years of the child's life. This design capitalizes on the approximately 10-fold increased chance of recurrence of ASD to siblings, in order to enable prospective investigation of ASD and its potential associated factors (Newschaffer et al., 2012). Women were recruited from four sites: Drexel/CHOP, Johns Hopkins University, UC Davis, and Northern CA Kaiser Permanente. To be included in EARLI, women had to be 18 years or older, less than 29 weeks pregnant, English or Spanish speaking, and living no farther than 2 h away from recruitment sites. Overall, 256 women were enrolled in EARLI and children of EARLI participants were enrolled between 2009 and 2012. The EARLI Study was reviewed and approved by the Drexel University Institutional Review Board, and all EARLI Study sites obtained local IRB approvals. Furthermore, all participants in the EARLI Study provided consent prior to study participation.

Participants were excluded from this study if they did not have outcome information available (either Social Responsiveness Scale (SRS) or diagnostic evaluation, both conducted at the 36-month visit) (n = 51) and if they did not have information on prenatal diet (n = 43). Twins (n = 4 pairs) were excluded due to the small number and correlation in exposures. Following these exclusions (-Figure S1), 154 women were included in our analytic sample.

## **Dietary information**

Dietary information was collected through a modified version of the National Cancer Institute's Dietary History Questionnaire (DHQ), which focused on diet since becoming pregnant and included additional questions from the validated version in order to capture contaminant exposure. This questionnaire was administered to EARLI participants twice during their pregnancy, at approximately 20 and 36 weeks gestation, and covered diet during the first and second half of pregnancy respectively. DHQ data from the first half of pregnancy were used for analyses here as a greater proportion of participants completed this questionnaire. The DHQ has been validated in previous work (Subar et al., 2001). The DHQ is comparable to the Willett/Harvard Food Frequency Questionnaire (FFQ) for which the PRBS was initially developed, as both questionnaires assess the main components of the diet, and notably, capture the same fruit and vegetable intake required to estimate the PRBS (Fawzi et al., 2004; Subar et al., 2001; W. C. Willett et al., 1985). In the DHQ administered to EARLI participants, organic consumption of individual fruits and vegetables was also assessed via ordinal response options ranging from almost always or always (approximately 90% of the time) to rarely or never (<10% of the time).

#### **PRBS** development and derivations

Previous studies have developed the PRBS as a novel approach to estimate dietary pesticide exposure based on FFQ data (Chiu et al., 2016; Chiu, Williams, Mínguez-Alarcón, Gillman, et al., 2018; Hu et al., 2016). This score aims to rank individual's pesticide exposure through fruit and vegetable intake, using data from the USDA's Pesticide Data Program (PDP). The PDP annually tests US agricultural commodities, following washing and peeling to mimic consumer behavior, for approximately 450 different pesticide residues (Pesticide Data Program Annual Summary, Calendar Year 2013, 2014). Data from the PDP was used to classify fruits and vegetables as high or low residue, according to three indexes assessing percent of samples with: 1) detectable pesticides, 2) above pesticide tolerances, and 3) >3 types of detectable pesticides. Across each of the indexes, foods were ranked into tertiles and assigned a score ranging from 0 (bottom tertile) to 2 (top tertile). Scores were summed to produce a pesticide rank score (PRBS) ranging from 0 (least contaminated) to 6 (most contaminated) (Table S1). Fruits and vegetables with scores  $\geq 4$  were considered high residue (HR), while those below 4 were considered low residue (LR). We calculated residue scores for foods tested by the PDP in the years covering EARLI study enrollment (2008-2013; flanking years were included to capture foods not tested during enrollment years). Since not all commodities are tested each year, foods not tested by the PDP during EARLI study enrollment were given a residue score from a previous study that used PDP data between 2006 and 2015 (13%) of foods included in the PRBS) (Chiu, Williams, Gillman, Gaskins, et al., 2018). Several residue scores were examined in this study.

# High and low pesticide residues and fruit and vegetable intake

For primary analyses, following prior work (Chiu et al., 2016; Chiu, Williams, Gillman, Gaskins, et al., 2018), we stratified foods according to high (residue score as defined  $\geq$ 4) or low (score <4) residues, and examined associations with consumption of these high-residue fruits and vegetables (HR-FV) and low-residue fruits and vegetables (LR-FV) respectively. Stratification by high and low residues will aim to categorize the presence of pesticides on fruits and vegetables. In addition to examining the pesticide residues in fruits and vegetables (as our primary question of interest), we also examined associations between overall fruit and vegetable (FV) intake with ASD outcomes.

# Weighted pesticide residue scores and modifications of the PRBS

Weighted residue scores were also calculated following published methods (Hu et al., 2016), by multiplying each fruit and vegetable's residue score by the individual's consumption frequency (servings/day) of that food reported in the DHQ, and then summing across all fruit and vegetables. Thus, this score aims to describe overall pesticide exposure in the diet. Class-specific organochlorine and organophosphate (OC-PRBS and OP-PRBS) scores were developed using methods discussed previously, including only OC or OP specific residues measured by the PDP (Hu et al., 2016). In addition to these scores following published strategies, we tested several modifications to the PRBS. Given fatty foods are a known source of OC pesticides due to their lipophilic nature (Fourth National Report on Human Exposure to Environmental Chemicals, 2009), we created a modified PRBS that incorporated available fatty foods from the PDP for 2008–2013 (meat, fish, milk, eggs) (PRBS-F). Next, we also incorporated organic food status into the PRBS by multiplying consumption frequency of each food item by the proportion of reported non-organic intake for that food (thereby reducing the estimated pesticide residue exposure; PRBS-organic). Across all residue scores examined, we parameterized the scores in quartiles to examine associations with outcomes under study; such categorization has also been shown to reduce potential misclassification of reported dietary data (W. Willett, 2013).

#### Outcome assessments

## ASD-related traits (primary outcome)

Quantitative ASD phenotype in the child was captured by the original preschool version of the SRS (Constantino & Gruber, 2005, 2012), administered to mother participants when the child was 36 months (Miller et al., 2021; Sacrey et al., 2021; Wetherby et al., 2021; Zwaigenbaum et al., 2015; Zwaigenbaum et al., 2019). The SRS generates a total raw score ranging from 0 to 195, where higher values indicate greater expression of the ASD-related phenotype. The SRS has been previously validated against the Autism Diagnostic Interview-Revised (ADI-R) (Constantino et al., 2003; Constantino et al., 2009), with coefficients ranging between 0.65 and 0.91. The SRS has previously shown strong psychometric properties, including high internal validity, reliability, reproducibility, and score stability over time (Constantino et al., 2003; Constantino et al., 2009), including in the general population (Constantino & Todd, 2003), and in ASD groups (Constantino et al., 2006). SRS scores are continuously distributed and have previously demonstrated evidence for heritability (Constantino & Todd, 2000, 2005).

## Cognitive development (secondary outcome)

Due to prior evidence for associations between pesticides and cognitive outcomes (Bouchard et al., 2010; Engel et al., 2016; Eskenazi et al., 2007; Furlong et al., 2017; Korrick & Sagiv, 2008; Rosas & Eskenazi, 2008), (Engel et al., 2016; Eskenazi et al., 2007; Furlong et al., 2017; Korrick & Sagiv, 2008), and the common co-occurrence of intellectual disability and ASD (with approximately 30%) of those with an ASD diagnosis also meeting criteria for intellectual disability) (Christensen et al., 2018; Lyall, Croen, Daniels, et al., 2017), we also examined associations with cognitive scores for comparison to associations with ASD-related traits. Cognitive development was assessed at 36 months using the Mullen Scales of Early Learning (MSEL) early learning composite (ELC) score (Mullen, 1995). This score is a standard cognitive tool that captures overall development, incorporating fine motor skills, visual reception, and both receptive and expressive language (Farmer et al., 2016), with prior evidence supporting internal reliability and validity against other scores that measure cognitive ability in children (Bishop et al., 2011; Farmer et al., 2016). Lower scores (mean ELC = 100; SD = 10) are indicative of poorer intellectual functioning (Bishop et al., 2011; Maski et al., 2011).

## Statistical analysis

Descriptive statistics of baseline characteristics were examined across the study sample and by quartiles of the PRBS. We qualitatively compared differences in baseline characteristics by examining the variance for continuous variables (via SD) and frequency of response for categorical variables across highest and lowest quartiles of exposure categories (Q1 HR-FV and Q4 HR-FV. Correlation between our exposure measurements was assessed using Spearman correlation coefficients. Due to minor differences between our FFQ used in EARLI and the FFQ used in the original PRBS development, we compared foods identified as high residue according to the published PRBS and our PRBS based on PDP data from 2008 to 2013.

For analyses examining diet-based pesticide exposure in association with SRS scores, we ranked individuals into quartiles based on their PRBS, with the lowest quartile as the referent. Linear regression was used to assess the relationship between quartiles of HR/LR-FV intake, FV intake, and weighted-PRBS. Adjusted models included the following covariates, selected based on a priori associations between exposures and outcomes: maternal age, prepregnancy BMI (weight in kg/height in m<sup>2</sup>), sex of the child, and total calorie intake (as a continuous variable in models examining dietary exposures) (Getz et al., 2016; Lyall, Croen, Daniels, et al., 2017; Shelton et al., 2010). In order to consider the role of healthy diet overall, we additionally adjusted for a modified version of the Alternative Healthy Eating Index (AHEI) dietary pattern (McCullough et al., 2002). This dietary pattern emphasizes a diet high in produce, whole grains, nuts, and polyunsaturated fatty acids, moderate consumption of alcohol, and low in red meat, trans fats, and saturated fatty acids. We excluded fruit and vegetable intake in the AHEI so FVs were not included twice in the model, and alcohol consumption was excluded due to contraindication in pregnancy (and near non-use in our population). Finally, as performed in prior work, HR-FV and LR-FV intake were adjusted for one another (Chiu et al., 2016; Chiu, Williams, Gillman, Gaskins, et al., 2018). Several other covariates tested for adjustment on the basis of potential or known a priori associations were: maternal prenatal vitamin use, a binary indicator for organic food consumption based on reported intake from the DHQ (as a complement to the PRBS-organic measure described above), study site, maternal education, race/ethnicity, household income, as well as folate and vitamin D intake. These additional factors were not included in final models as they did not change estimates by >10% (and given inclusion of the latter two nutrients may ultimately complicate the interpretation of FV effects). Covariate missingness was low (5 individuals or fewer, or roughly 3% or less), and simple imputation was used for individuals missing covariate information. Analyses of secondary outcome measure (MSEL ELC scores) followed the approach outlined for SRS scores. Across analyses, estimates with 95% confidence intervals not including the null value of 0 (consistent with p < 0.05) were interpreted as statistically significant. All analyses were conducted using SAS 9.4 software.

## RESULTS

Characteristics of the 154 EARLI participants included in primary analyses are shown in Table 1. East coast participants, Hispanic, Black and multi-racial mothers and those with less than a college degree were more likely to consume more high pesticide residue fruit and vegetables (HR-FV) than non-Hispanic or white mothers. Individuals who were ranked in the top quartile of HR-FV had a lower mean pre-pregnancy BMI compared to those in the bottom quartile of HR-FV (Table 1).

Dietary characteristics of those in the top and bottom quartile of HR-FV intake are shown in Table S2. As expected, those in the highest quartile of the HR-FV had higher fruit and vegetable intake. Overall, fruit and vegetable intake were highly correlated with one another ( $\rho = 0.67$ ), as were HR and LR-FV intake ( $\rho = 0.65$ ). Those in the top quartile of HR-FV intake reported higher intake of organic foods (18.4% vs. 2.6%) and had higher intake of several nutrients of relevance to (and with prior evidence for inverse associations with) ASD compared to those in the bottom quartile of HR-FV (Table S2).

Pesticide residue scores derived over EARLI study years (2008–2013) were comparable (the same or within 1 rank) across 88% of foods in the previously published PRBS based on PDP data covering 2006–2015 (Chiu, Williams, Gillman, Gaskins, et al., 2018). In addition, there was 85% similarity of foods identified as high residue between our score and the published PRBS, providing support for comparability and reliability of our PRBS. In testing modifications to the PRBS, fatty foods available from the PDP (beef, fish, milk & eggs, which were incorporated in our PRBS-fatty foods) were all low residue (residue scores <4). Consistent intake of organic foods was relatively low in our study population (n = 17classified as regular organic consumers, defined according to report of eating organic foods  $\geq 50\%$  of the time). Though we did not examine FV intake based on dietary data from the second half of pregnancy, among the subset with DHQ data across two time points (approximately 50% of the eligible participants), the correlation of food intake that contributed to the PRBS was high ( $\rho = 0.75$ ).

#### Associations with pesticide residues

In analyses of HR/LR-FV intake with ASD-related traits, overall, we did not observe strong or statistically significant associations (Table 2; beta estimates consistent with approximately a 2-point increase and 5-point decrease in SRS scores, with 95% CIs including the null, were found for Q4 vs. Q1 of HR-FV and LR-FV, respectively). After adjustment, increased HR/LR-FV intake was not associated with changes in SRS scores (Table 2). When examining associations of fruit and vegetable intake, vegetable intake, as well as combined fruit/vegetable intake, were associated with decreases in SRS scores of approximately 8 to 12 points, which did not reach statistical significance (Q4 vs. Q1 adjusted  $\beta = -8.70, 95\%$  CI -23.9, 6.5 for vegetable intake which

**TABLE 1** Basic characteristics of the study population overall and by lowest and highest high-residue fruit and vegetable intake (HR-FV) quartile

	Overall $(n = 154)$	HR-FV Q1 ( $n = 38$ )	HR-FV Q4 (n = 38)
		n (%)	
Study sites <sup>a</sup>			
West Coast	77 (50%)	22 (57.9%)	17 (44.7%)
East Coast	77 (50%)	16 (42.1%)	21 (55.3%)
Child sex			
Male	83 (53.9%)	25 (65.8%)	20 (52.6%)
Female	71 (46.1%)	13 (34.2%)	18 (47.4%)
Maternal ethnicity			
Hispanic/Latino	30 (19.5%)	5 (13.2%)	9 (23.7%)
Not Hispanic/Latino	124 (80.5%)	33 (86.8%)	29 (76.3%)
Maternal race			
White	104 (67.5%)	25 (65.8%)	20 (52.6%)
Black/African American	9 (5.8%)	2 (5.3%)	5 (13.2%)
Asian & Pacific Islander	20 (13.0%)	7 (18.4%)	6 (15.8%)
Multiple/Other Race	16 (10.4%)	3 (7.9%)	5 (13.2%)
Missing	5 (3.3%)	1 (2.6%)	2 (5.3%)
Household income			
\$0-50,000	36 (23.4%)	10 (26.3%)	9 (23.7%)
\$50,001-100,000	58 (37.7%)	15 (39.5%)	15 (39.5%)
\$100,001+	60 (38.9%)	13 (34.2%)	14 (36.8%)
Maternal education level			
High School Diploma or Less	20 (13.0%)	5 (13.2%)	6 (15.8%)
Some College	43 (27.9%)	10 (26.3%)	13 (34.2%)
Bachelor's degree	46 (29.9%)	10 (26.3%)	9 (23.7%)
Graduate degree	44 (28.6%)	13 (34.2%)	9 (23.7%)
Missing	1 (0.6%)		1 (2.6%)
Prenatal smoking			
Active smoking	5 (3.3%)	3 (7.9%)	0 (0%)
Secondhand, not active smoking	2 (1.3%)	1 (2.6%)	0 (0%)
No smoking exposure	120 (77.9%)	26 (68.4%)	31 (81.6%)
Missing	27 (17.5%)	8 (21.1%)	7 (18.4%)
Maternal weight			
Obese	47 (30.5%)	14 (36.8%)	7 (18.4%)
Overweight	40 (26.0%)	8 (21.1%)	12 (31.6%)
Normal	61 (39.6%)	16 (42.1%)	17 (44.7%)
Underweight	2 (1.3%)	0 (0%)	1 (2.6%)
Missing	4 (2.6%)		1 (2.6%)
Parity			
1	68 (44.2%)	16 (42.1%)	15 (29.5%)
2	62 (40.3%)	15 (39.5%)	19 (50.0%)
3+	23 (14.9%)	7 (18.4%)	3 (7.9%)
Missing	1 (0.6%)		1 (2.6%)
Prenatal vitamin use			
Yes	145 (94.2%)	36 (94.7%)	37 (97.4%)
No	8 (5.2%)	2 (5.3%)	0 (0%)
Missing	1 (0.6%)		1 (2.6%)
-	× /		(Continues

#### TABLE 1 (Continued)

	Overall $(n = 154)$	HR-FV Q1 ( $n = 38$ )	HR-FV Q4 ( $n = 38$ )
	n (%)		
Autism diagnosis			
Yes	32 (20.8%)	13 (34.2%)	8 (21.1%)
No	120 (77.9%)	25 (65.8%)	30 (78.9%)
Missing	2 (1.3%)		
		Mean (std)	
Maternal age (years)	33.9 (4.6)	33.6 (3.8)	33.7 (5.1)
Pre-pregnancy BMI (kg/m <sup>2</sup> ) <sup>b</sup>	28.0 (7.1)	28.8 (7.2)	26.1 (5.1)
Total SRS raw score <sup>b</sup>	35.8 (26.2)	38.0 (27.5)	33.5 (29.2)
Total SRS T score <sup>b</sup>	47.7 (10.1)	48.6 (10.7)	46.9 (11.3)
Mullen ELC score <sup>b</sup>	99.1 (20.9)	96.0 (20.8)	99.1 (23.3)

<sup>a</sup>EARLI recruited participants at four sites; West coast study sites include UC Davis and Kaiser Permanente Northern CA; East coast study sites include Drexel/CHOP and Johns Hopkins University.

<sup>b</sup>Four were missing information on BMI. Twenty-three individuals in EARLI did not have SRS scores and 9 did not have Mullen Scores (none overlapping).

is consistent with roughly a 1/3 SD unit decrease in raw SRS scores; Q4 vs. Q1 adjusted  $\beta = -12.76$ , 95% CI: -27.8, 2.3 for combined fruit and vegetable intake) (Table 2). Fruit intake on its own did not suggest the same association (Q4 vs. Q1 adjusted  $\beta$  for SRS = -0.11, 95% CI -14.1, 13.8 for fruit intake) (Table 2).

When examining associations with weighted residue scores, as for HR/LV-FV intake, we did not find evidence for associations with ASD-related traits (Table 3). However, a reduction in SRS scores was observed for the PRBS incorporating reported organic intake (PRBSorganic) (Q4 vs. Q1 adjusted  $\beta = -16.86$ , 95% CI -31.9, -1.8) (Table 3). Additional adjustment for a range of other factors, including maternal folate intake and additional maternal dietary and demographic characteristics, did not materially alter these findings for SRS scores (Table S3).

In secondary analyses, pesticide residue scores were not associated with MSEL ELC scores (Tables S4 and S5). However, consistent with associations with SRS scores, combined fruit and vegetable intake was associated with higher MSEL ELC scores (indicative of improved cognitive development), driven primarily by vegetable intake (Table S4).

## DISCUSSION

In this high-familial probability cohort, we examined the relationship between pesticide residues in the diet during pregnancy in association with child ASD-related outcomes at age 3. Overall, we did not find evidence to support associations in the hypothesized positive direction with ASD-related traits, and in some cases, associations with diet-based residue in the opposite direction

(suggesting an inverse relationship) were found. However, decreases in ASD-related traits were observed with overall fruit and vegetable intake, suggesting beneficial nutrients in these foods could outweigh adverse effects of pesticide residues in the diet on these outcomes. Additional considerations may play into these findings, as further discussed below.

The FFQ used in our study, a modified version of the DHQ, was shown to be comparable to the Harvard/ Willett FFQ used in original work for assessment of dietary pesticide exposure. Score development using the PDP was reproducible across different study years, and there were consistencies in pesticide residues across foods assessed in these studies. Original work developing the PRBS verified the score's utility in approximating urinary and serum pesticide biomarkers in a large populationbased cohort, validating the score is an accurate approximation of total pesticide exposure (Hu et al., 2016).

As noted, we had hypothesized that pesticide residue scores would be associated with increases in ASD-related traits, given prior evidence for adverse effects of prenatal exposure to pesticides on neurodevelopment (Korrick & Sagiv, 2008; Rosas & Eskenazi, 2008). However, null associations and results consistent with a non-significant decrease in ASD-related traits were observed in many analyses. Results for our primary exposure assessment, intake of high and low residue fruits and vegetables, did not suggest strong associations with ASD-related traits. Given the high correlation between high and low residue FV, future work should further consider ways to address the challenge of identifying distinct contributions of these correlated exposure sources, such as in mixtures analyses. Results for associations with ASD-related traits were also similar for the weighted residue scores incorporating FV intake more broadly. We observed an association

**TABLE 2** Crude and adjusted associations ( $\beta$  estimates and 95% confidence intervals) between maternal HR/LR-FV intake & overall fruit and vegetable intake during pregnancy and child SRS scores (n = 131)

Associations with SRS scores					
	n	Crude (β, 95% CI)	Adjusted (β, 95% CI) <sup>a</sup>		
HR-FV	HR-FV				
Q1	34	(Referent)	(Referent)		
Q2	33	-2.67 (-15.1, 9.8)	0.10 (-12.2, 12.4)		
Q3	34	-2.09 (-14.5, 10.3)	4.34 (-9.9, 18.6)		
Q4	30	-4.47 (-17.2, 8.3)	1.37 (-14.6, 17.4)		
LR-FV					
Q1	32	(Referent)	(Referent)		
Q2	32	-8.53 (-21.1, 4.1)	-4.45 (-17.7, 8.8)		
Q3	36	2.22 (-10.0, 14.5)	6.01 (-8.5, 20.5)		
Q4	31	-6.40 (-19.1, 6.3)	-5.57 (-22.4, 11.3)		
Fruit intake					
Q1	31	(Referent)	(Referent)		
Q2	35	-0.46 (-13.1, 12.1)	1.71 (-10.4, 13.8)		
Q3	34	-0.65 (-13.3, 12.0)	2.95 (-9.6, 15.5)		
Q4	31	-2.90 (-15.9, 10.1)	-0.11 (-14.1, 13.8)		
Vegetable intake					
Q1	33	(Referent)	(Referent)		
Q2	33	-9.79 (-22.2, 2.7)	-8.56 (-20.8, 3.7)		
Q3	33	-5.00 (-17.4, 7.4)	-0.78 (-14.4, 12.9)		
Q4	32	-8.79 (-21.3, 3.7)	-8.70 (-23.9, 6.5)		
Total fruit and vegetable intake					
Q1	33	(Referent)	(Referent)		
Q2	33	-16.58 (-28.7, -4.4)	-11.28 (-23.4, 0.83)		
Q3	34	-4.48 (-16.6, 7.6)	-0.45 (-13.3, 12.4)		
Q4	31	-13.1 (-25.5, -0.8)	-12.76 (-27.8, 2.3)		

<sup>a</sup>Adjusted for maternal age, maternal pre-pregnancy BMI, child sex, total calorie intake, and a modified version of the AHEI (Alternative Healthy Eating Index) dietary pattern score that excluded fruits, vegetables, and alcohol. HR-FV and LR-FV additionally adjusted for one another.

bordering statistical significance of higher fruit and vegetable intake and decreases in ASD-related traits, which appeared to be driven more by vegetable intake. Taken together, these findings may suggest that the potential effects of pesticide residues on ASD-related outcomes may be counterbalanced by the effects of vegetable intake. Further work should examine this hypothesis, address factors that may drive the suggested protective association of vegetable intake based on nutrients or other health-related factors, and replicate this finding in larger samples.

We also did not observe differences in associations with outcomes according to the modified versions of the PRBS we tested, with the potential exception (in the unexpected direction) of the organic PRBS. Although we had a small number of women reporting regular intake of organic foods, incorporating reported organic intake into

**TABLE 3** Crude and adjusted associations ( $\beta$  estimates and 95% confidence intervals) between weighted versions of maternal pesticide residue burden scores during pregnancy and child raw total SRS scores (n = 131)

	n	Crude (β, 95% CI)	Adjusted (β, 95% CI) <sup>a</sup>	
PRBS				
Q1	34	(Referent)	(Referent)	
Q2	32	-1.83 (-14.4, 10.7)	2.66 (-10.4, 15.8)	
Q3	35	-3.49 (-15.8, 8.8)	2.98 (-10.4, 16.4)	
Q4	30	-4.51 (-17.3, 8.3)	0.28 (-16.0, 16.5)	
PRBS-	F			
Q1	33	(Referent)	(Referent)	
Q2	33	-3.39 (-15.9, 9.2)	1.31 (-11.6, 14.2)	
Q3	35	-3.82 (-16.2, 8.5)	2.87 (-10.8, 16.6)	
Q4	30	-5.15 (-18.0, 7.7)	-0.28 (-16.7, 16.1)	
PRBS-Organic				
Q1	32	(Referent)	(Referent)	
Q2	35	-5.38 (-17.7, 7.0)	-5.79 (-17.7, 6.1)	
Q3	33	-3.51 (-16.0, 9.0)	-4.11 (-16.9, 8.7)	
Q4	31	-11.39 (-24.1, 1.3)	-16.86 (-31.9, -1.8)	
OC-PRBS				
Q1	33	(Referent)	(Referent)	
Q2	33	-3.43 (-16.0, 9.1)	0.56 (-12.1, 13.3)	
Q3	33	-1.15 (-13.7, 11.4)	5.06 (-8.4, 18.5)	
Q4	32	-4.28 (-16.9, 8.4)	-0.69 (-16.4, 15.1)	
OP-PRBS				
Q1	34	(Referent)	(Referent)	
Q2	32	-2.53 (-15.1, 10.0)	2.07 (-10.4, 14.6)	
Q3	35	-2.20 (-14.5, 10.1)	3.76 (-10.0, 17.6)	
Q4	30	-4.89 (-17.7, 7.9)	0.87 (-15.4, 17.1)	

<sup>a</sup>Adjusted for maternal age, maternal pre-pregnancy BMI, child sex, total calorie intake, and a modified version of the AHEI (Alternative Healthy Eating Index) dietary pattern score that excluded fruits, vegetables, and alcohol.

scores did result in improved relationships between score ranks (PRBS-organic). However, accounting for organic intake in this modified version of the PRBS may have also unintentionally captured other factors related to healthy behaviors, and may have introduced negative bias into this score when examining associations with outcomes.

Though adjustment for additional sociodemographic and dietary factors did not materially alter our results, we cannot rule out potential residual confounding by more challenging-to-capture health-related characteristics. Those who were ranked in the top quartile of HR-FV intake in our study had a lower BMI, ate a generally healthier diet, and more often ate organic foods compared to those in the bottom quartile; factors consistent with characteristics of a previous study that assessed high-residue fruit and vegetable intake in a cohort of women receiving infertility treatment (Chiu, Williams, Gillman, Gaskins, et al., 2018). For ASD, previous studies have demonstrated decreased risk with maternal folic acid supplementation, as well as (with somewhat less consistency) polyunsaturated fatty acids (DeVilbiss et al., 2015; Julvez et al., 2016; Lyall et al., 2013; Schmidt et al., 2012). Nonetheless, in addition to residual confounding, it is also possible that live birth bias could also play into unexpected null and inverse results (Raz et al., 2018), given that prior work has shown that women who were in the top quartile of high-residue pesticide exposure from fruit and vegetable intake had a greater probability of pregnancy loss and a lower probability of clinical pregnancy compared to women who were in the bottom quartile, representing the lowest exposure to pesticide residues (Chiu, Williams, Gillman, Gaskins, et al., 2018).

The residue score accounts for hundreds of pesticides tested in the PDP across several classes. As such, it represents an efficient way to capture exposure to multiple classes of pesticides and may help address pesticide mixture effects. Few prior studies have considered combined effects of pesticides on ASD-related outcomes, though a number of prior studies have supported associations between prenatal pesticide exposure and ASD outcomes. Braun and colleagues examined the relationship between environmental chemical exposures and ASD-related traits in mixture analyses in the HOME cohort in Ohio, and found measured levels of trans-nonachlor (accounting for other pesticides and endocrine-disrupting chemicals) were associated with increases in SRS-T scores (Braun et al., 2014). In a large nested case-control study of births in Finland conducted by Brown et al, children in the top 75th percentile of maternal  $p_{,p'}$ -DDE levels increased the odds of ASD (Brown et al., 2018). The Early Markers for Autism (EMA) population-based case control study conducted in California reported null results when assessing the relationship between p,p'-DDE and ASD diagnoses in California, but showed a nonmonotonic association with intellectual disability (Lyall, Croen, Sjödin, et al., 2017).

Though fruits and vegetables are the primary source of exposure to pesticides in use today, intake of these foods may not best capture total dietary exposure to some classes of pesticides including OCs, given OC pesticides are lipophilic chemicals (Fourth National Report on Human Exposure to Environmental Chemicals, 2009). OC pesticides have been largely banned and replaced by the OP and pyrethroid classes. Previous work on sources of pesticide exposures in diet found that parents and children consumed most OC pesticides (including p,p'-DDE as well as others) from dairy, and fish (Vogt et al., 2012). Furthermore, prior work examining and developing classspecific PRBS has reported stronger relationships between the OP-PRBS and measured levels of OP metabolites (creatinine-adjusted dimethylphosphate, diethylphosphate, dimethylthiophosphate, diethylthiophosphate, and dimethyldithiophosphate) than between the OC-PRBS and measured levels of OC metabolites (lipid-adjusted

hexachlorobenzene, oxychlordane, *trans*-nonachlor and heptachlor epoxide) (Hu et al., 2016). This could be in part explained by reduced body burden of OCs in the decades following their ban (Hagmar et al., 2006). Though the fatty foods included here were classified as low residue by the PDP, future work addressing pesticide exposure may consider additional food sources.

Different classes of pesticides may have differing associations with ASD, and there is evidence from other studies pesticides such as OPs and pyrethroids may be related to ASD outcomes (Shelton et al., 2014; von Ehrenstein et al., 2019). OPs are known to inhibit acetylcholinesterase in the brain, resulting in neurotoxic effects at high exposure (Fourth National Report on Human Exposure to Environmental Chemicals, 2009; Sagiv et al., 2018). However, neurological and developmental effects of routine low-dose exposure to OPs in humans is not well categorized. A 10-fold increase in prenatal urinary OP levels (specifically DAPs and DMs) was positively associated with child SRS scores for women who live close to agricultural applications during pregnancy (Sagiv et al., 2018). At the same time, other studies have shown null results between OP pesticides and SRS scores (Millenson et al., 2017). Another high familial probability cohort study in California (MARBLES: Markers of Autism Risk in Babies-Learning Early Signs), which enrolled participants at a similar time to EARLI, showed no association between urinary OP markers and ASD diagnosis (Philippat et al., 2018). These mixed findings suggest the need for further work examining associations with measured levels of pesticides across multiple classes and comparing these associations with dietary-based residue exposures to better consider the role of primary sources. In addition, given suggestions that combined exposure can create measurable risk even when the individual toxicant is not associated with adverse outcomes (Chiu, Bellavia, et al., 2018; Kortenkamp, 2007; Rider et al., 2010), studies with the ability to examine multiple classes of pesticides will be useful in further addressing the question of combined effects.

In this study, we had the ability to examine associations with broader cognitive development. Mullen scores were generally reduced with higher PRBS quartiles, although these associations were not statistically significant. Other studies have also examined more general neurodevelopmental outcomes that may be influenced by pesticides (Bradman et al., 2003; Dewan et al., 2013). Several studies have shown that prenatal exposure to OPs are related to cognitive deficits (Bouchard et al., 2011; Engel et al., 2016; Eskenazi et al., 2007; Philippat et al., 2018; Rauh et al., 2012) and ASD-related traits (Eskenazi et al., 2007; Furlong et al., 2014; Sagiv et al., 2018). However, there are inconsistencies regarding associations between OC exposure and child neurodevelopmental outcomes (Eskenazi et al., 2006; Gaspar et al., 2015; Lyall, Croen, Sjödin, et al., 2017; Yamazaki

et al., 2018), and some studies have reported sex differences (Braun et al., 2014; Gaspar et al., 2015; Pan et al., 2009; Sioen et al., 2013), which we did not have sufficient power to address here. Several mechanisms may underlie these potential associations. Thyroid hormone disruption, inflammation, and oxidative stress have the ability to influence neurodevelopment of the fetus (and have been linked with ASD [Garay & McAllister, 2010; Goines & Van de Water, 2010; Lyall, Anderson, Kharrazi, & Windham, 2017; Soldin et al., 2003; Yau et al., 2015]) and have been shown to be influenced by pesticide exposure (De Felice et al., 2016; C. Li et al., 2014; Mense et al., 2006; Starek-&wiechowicz et al., 2017; Wang et al., 2017).

This is the first study to address pesticide residues in the diet in association with ASD-related outcomes. Strengths of our work include use of prospectively collected data, notably dietary data during pregnancy, and the consideration of several versions of the weighted-PRBS. These included the incorporation of organic food status and inclusion of fatty foods from the PDP, to help address other sources of dietary exposure to pesticides in addition to conventional produce.

Several limitations should also be noted. Though fruits and vegetables are the main source of pesticide exposure in the diet, we were limited in our ability to address residues from other food sources due to the scope of the PDP database. The majority of foods tested by the PDP were fruits and vegetables (Pesticide Data Program Annual Summary, Calendar Year 2013, 2014), and consequently there were limited fatty foods, such as fish and dairy, tested by the PDP that were relevant to commodities assessed in the DHQ. Dietary data from EARLI was collected using a slightly modified version of the DHQ, and although the DHQ has been validated and the changes made did not alter calculation of the PRBS, the modified version of the DHQ used here has not itself been examined in validation work. Furthermore, we cannot rule out potential exposure misclassification due to self-report of diet, however our work and prior studies have demonstrated broad validity not only of PRBS but also of self-reported diet through FFQ measurement (Rimm et al., 1992; Subar et al., 2001), and any such misclassification would be expected to be non-differential with respect to the outcome due to prospective collection of data. Finally, we also had a small sample size, limiting our ability to examine autism diagnosis due to an insufficient number of cases in our study, and yielding wide confidence intervals in several analyses, highlighting the need for future larger studies on this topic.

## CONCLUSIONS

Diet is not only the primary source of pesticide exposure in the general population (Fourth National Report on Human Exposure to Environmental Chemicals, 2009; Morgan, 2012; Xue et al., 2014), but it is also a source of micronutrients that can potentially reduce risk of adverse outcomes or interact with pesticide exposure. Additional work is needed to parse the role of diet and pesticides in ASD and related outcomes. We did not find associations indicating an increased probability of ASD-related traits with dietary pesticide residue scores in this high-familial probability cohort. Instead, we found that vegetable intake, or perhaps key nutrients in vegetables or correlates of a diet with high vegetable intake, was inversely associated with ASD-related traits. Future work should follow up on the suggestive findings we observed for this association with vegetable intake, and further consider combined effects of prenatal exposure to pesticides on ASD-related outcomes.

## ETHICS STATEMENT

The EARLI Study was reviewed and approved by the Drexel University Institutional Review Board (Project no. 71109; Protocol no. 17862), and all EARLI Study sites obtained local IRB approvals.

## ACKNOWLEDGMENTS

The EARLI study was funded by the National Institute of Environmental Health Sciences, the National Institute of Mental Health, the National Institute of Child Health and Human Development, and the National Institute of Neurologic Disease and Stroke (R01 ES016443), with additional funding from Autism Speaks (AS 5938). This work was also supported by pilot funding from the Drexel University Department of Epidemiology. The authors would like to thank EARLI study participants. In addition, we thank Elizabeth Kauffman for her contributions to data management.

#### **CONFLICT OF INTEREST**

The authors declare no potential conflict of interest.

### DATA AVAILABILITY STATEMENT

Data are available upon reasonable request to the authors. In addition, EARLI participates in sharing data through the National Database for Autism Research (NDAR).

#### ORCID

*Emily E. Joyce* https://orcid.org/0000-0001-9722-5596 *Lisa A. Croen* https://orcid.org/0000-0001-7849-9428 *Irva Hertz-Picciotto* https://orcid.org/0000-0001-6952-2390

*Rebecca J. Schmidt* https://orcid.org/0000-0003-1582-2747

Heather Volk https://orcid.org/0000-0002-5415-2494 Kristen Lyall https://orcid.org/0000-0002-4633-0799

### REFERENCES

Barkoski, J. M., Philippat, C., Tancredi, D., Schmidt, R. J., Ozonoff, S., Barr, D. B., Elms, W., Bennett, D. H., & Hertz-Picciotto, I. (2021). In utero pyrethroid pesticide exposure in relation to autism spectrum disorder (ASD) and other neurodevelopmental outcomes at 3 years in the MARBLES longitudinal cohort. *Environmental Research*, 194, 110495. https://doi.org/10.1016/j.envres.2020.110495

- Bishop, S. L., Guthrie, W., Coffing, M., & Lord, C. (2011). Convergent validity of the Mullen Scales of Early Learning and the differential ability scales in children with autism spectrum disorders. *American Journal on Intellectual and Developmental Disabilities*, 116(5), 331– 343. https://doi.org/10.1352/1944-7558-116.5.331
- Bouchard, M. F., Bellinger, D. C., Wright, R. O., & Weisskopf, M. G. (2010). Attention-deficit/hyperactivity disorder and urinary metabolites of organophosphate pesticides. *Pediatrics*, 125(6), e1270– e1277. https://doi.org/10.1542/peds.2009-3058
- Bouchard, M. F., Chevrier, J., Harley, K. G., Kogut, K., Vedar, M., Calderon, N., Trujillo, C., Johnson, C., Bradman, A., Barr, D. B., & Eskenazi, B. (2011). Prenatal exposure to organophosphate pesticides and IQ in 7-year-old children. *Environmental Health Perspectives*, 119(8), 1189–1195. https://doi.org/10.1289/ ehp.1003185
- Bradman, A., Barr, D. B., Claus Henn, B. G., Drumheller, T., Curry, C., & Eskenazi, B. (2003). Measurement of pesticides and other toxicants in amniotic fluid as a potential biomarker of prenatal exposure: A validation study. *Environmental Health Perspectives*, 111(14), 1779–1782. https://doi.org/10.1289/ehp.6259
- Braun, J. M., Kalkbrenner, A. E., Just, A. C., Yolton, K., Calafat, A. M., Sjödin, A., Hauser, R., Webster, G. M., Chen, A., & Lanphear, B. P. (2014). Gestational exposure to endocrine-disrupting chemicals and reciprocal social, repetitive, and stereotypic behaviors in 4- and 5-year-old children: The HOME study. *Environmental Health Perspectives*, 122(5), 513– 520. https://doi.org/10.1289/ehp.1307261
- Brown, A. S., Cheslack-Postava, K., Rantakokko, P., Kiviranta, H., Hinkka-Yli-Salomäki, S., McKeague, I. W., Surcel, H. M., & Sourander, A. (2018). Association of Maternal Insecticide Levels With Autism in Offspring From a National Birth Cohort. *The American Journal of Psychiatry*, 175(11), 1094–1101. https://doi. org/10.1176/appi.ajp.2018.17101129
- Caspersen, I. H., Kvalem, H. E., Haugen, M., Brantsæter, A. L., Meltzer, H. M., Alexander, J., Thomsen, C., Frøshaug, M., Bremnes, N. M., Broadwell, S. L., Granum, B., Kogevinas, M., & Knutsen, H. K. (2016). Determinants of plasma PCB, brominated flame retardants, and organochlorine pesticides in pregnant women and 3 year old children in The Norwegian Mother and Child Cohort Study. *Environmental Research*, 146, 136–144. https://doi.org/10.1016/j.envres.2015.12.020
- Cequier, E., Sakhi, A. K., Haug, L. S., & Thomsen, C. (2017). Exposure to organophosphorus pesticides in Norwegian mothers and their children: Diurnal variability in concentrations of their biomarkers and associations with food consumption. *Science of the Total Environment*, 590-591, 655–662. https://doi.org/10.1016/j.scitotenv. 2017.03.017
- Chiu, Y. H., Bellavia, A., James-Todd, T., Correia, K. F., Valeri, L., Messerlian, C., Ford, J. B., Mínguez-Alarcón, L., Calafat, A. M., Hauser, R., Williams, P. L., & EARTH Study Team. (2018). Evaluating effects of prenatal exposure to phthalate mixtures on birth weight: A comparison of three statistical approaches. *Environment International*, *113*, 231–239. https://doi.org/10.1016/j.envint.2018. 02.005
- Chiu, Y. H., Gaskins, A. J., Williams, P. L., Mendiola, J., Jørgensen, N., Levine, H., Hauser, R., Swan, S. H., & Chavarro, J. E. (2016). Intake of fruits and vegetables with low-to-moderate pesticide residues is positively associated with semen-quality parameters among young healthy men. *The Journal of Nutrition*, 146(5), 1084–1092. https://doi.org/10.3945/jn.115.226563
- Chiu, Y. H., Williams, P. L., Gillman, M. W., Gaskins, A. J., Mínguez-Alarcón, L., Souter, I., Toth, T. L., Ford, J. B., Hauser, R., Chavarro, J. E., & EARTH Study Team. (2018). Association

between pesticide residue intake from consumption of fruits and vegetables and pregnancy outcomes among women undergoing infertility treatment with assisted reproductive technology. *JAMA Internal Medicine*, *178*(1), 17–26. https://doi.org/10.1001/jamainternmed.2017.5038

- Chiu, Y. H., Williams, P. L., Gillman, M. W., Hauser, R., Rifas-Shiman, S. L., Bellavia, A., Fleisch, A. F., Oken, E., & Chavarro, J. E. (2018). Maternal intake of pesticide residues from fruits and vegetables in relation to fetal growth. *Environment International*, *119*, 421–428. https://doi.org/10.1016/j.envint.2018. 07.014
- Chiu, Y. H., Williams, P. L., Mínguez-Alarcón, L., Gillman, M., Sun, Q., Ospina, M., Calafat, A. M., Hauser, R., & Chavarro, J. E. (2018). Comparison of questionnaire-based estimation of pesticide residue intake from fruits and vegetables with urinary concentrations of pesticide biomarkers. *Journal of Exposure Science & Environmental Epidemiology*, 28(1), 31–39. https:// doi.org/10.1038/jes.2017.22
- Christensen, D. L., Braun, K. V. N., Baio, J., Bilder, D., Charles, J., Constantino, J. N., Daniels, J., Durkin, M. S., Fitzgerald, R. T., Kurzius-Spencer, M., Lee, L. C., Pettygrove, S., Robinson, C., Schulz, E., Wells, C., Wingate, M. S., Zahorodny, W., & Yeargin-Allsopp, M. (2018). Prevalence and characteristics of autism spectrum disorder among children aged 8 years – Autism and Developmental Disabilities Monitoring Network, 11 Sites, United States, 2012. MMWR Surveillance Summaries, 65(13), 1–23. https://doi. org/10.15585/mmwr.ss6513a1
- Constantino, J. N., Abbacchi, A. M., Lavesser, P. D., Reed, H., Givens, L., Chiang, L., Gray, T., Gross, M., Zhang, Y., & Todd, R. D. (2009). Developmental course of autistic social impairment in males. *Development and Psychopathology*, 21(1), 127–138. https://doi.org/10.1017/s095457940900008x
- Constantino, J. N., Davis, S. A., Todd, R. D., Schindler, M. K., Gross, M. M., Brophy, S. L., Metzger, L. M., Shoushtari, C. S., Splinter, R., & Reich, W. (2003). Validation of a brief quantitative measure of autistic traits: Comparison of the social responsiveness scale with the autism diagnostic interview-revised. *Journal of Autism and Developmental Disorders*, 33(4), 427–433. https://doi. org/10.1023/a:1025014929212
- Constantino, J. N., & Gruber, C. (2005). Social Responsiveness Scale (SRS). Los Angeles, CA: Western Psychological Services.
- Constantino, J. N., & Gruber, C. (2012). Social Responsiveness Scale (2nd ed.). Western Psychological Services.
- Constantino, J. N., Lajonchere, C., Lutz, M., Gray, T., Abbacchi, A., McKenna, K., Singh, D., & Todd, R. D. (2006). Autistic social impairment in the siblings of children with pervasive developmental disorders. *The American Journal of Psychiatry*, 163(2), 294– 296. https://doi.org/10.1176/appi.ajp.163.2.294
- Constantino, J. N., & Todd, R. D. (2000). Genetic structure of reciprocal social behavior. *The American Journal of Psychiatry*, 157(12), 2043–2045. https://doi.org/10.1176/appi.ajp.157.12.2043
- Constantino, J. N., & Todd, R. D. (2003). Autistic traits in the general population: A twin study. *Archives of General Psychiatry*, 60(5), 524–530. https://doi.org/10.1001/archpsyc.60.5.524
- Constantino, J. N., & Todd, R. D. (2005). Intergenerational transmission of subthreshold autistic traits in the general population. *Biological Psychiatry*, 57(6), 655–660. https://doi.org/10.1016/j. biopsych.2004.12.014
- De Felice, A., Greco, A., Calamandrei, G., & Minghetti, L. (2016). Prenatal exposure to the organophosphate insecticide chlorpyrifos enhances brain oxidative stress and prostaglandin E2 synthesis in a mouse model of idiopathic autism. *Journal of Neuroinflammation*, 13(1), 149. https://doi.org/10.1186/s12974-016-0617-4
- DeVilbiss, E. A., Gardner, R. M., Newschaffer, C. J., & Lee, B. K. (2015). Maternal folate status as a risk factor for autism spectrum disorders: A review of existing evidence. *The British Journal of*

*Nutrition*, *114*(5), 663–672. https://doi.org/10.1017/ s0007114515002470

- Dewan, P., Jain, V., Gupta, P., & Banerjee, B. D. (2013). Organochlorine pesticide residues in maternal blood, cord blood, placenta, and breastmilk and their relation to birth size. *Chemosphere*, 90(5), 1704–1710. https://doi.org/10.1016/j.chemosphere.2012.09.083
- Diagnostic and statistical manual of mental disorders: DSM-V. (2013). Diagnostic and statistical manual of mental disorders. American Psychiatric Association.
- Engel, S. M., Bradman, A., Wolff, M. S., Rauh, V. A., Harley, K. G., Yang, J. H., Hoepner, L. A., Barr, D. B., Yolton, K., Vedar, M. G., Xu, Y., Hornung, R. W., Wetmur, J. G., Chen, J., Holland, N. T., Perera, F. P., Whyatt, R. M., Lanphear, B. P., & Eskenazi, B. (2016). Prenatal Organophosphorus Pesticide Exposure and Child Neurodevelopment at 24 Months: An Analysis of Four Birth Cohorts. *Environmental Health Perspectives*, 124(6), 822–830. https://doi.org/10.1289/ehp.1409474
- Eskenazi, B., Marks, A. R., Bradman, A., Fenster, L., Johnson, C., Barr, D. B., & Jewell, N. P. (2006). In utero exposure to dichlorodiphenyltrichloroethane (DDT) and dichlorodiphenyldichloroethylene (DDE) and neurodevelopment among young Mexican American children. *Pediatrics*, 118(1), 233–241. https://doi. org/10.1542/peds.2005-3117
- Eskenazi, B., Marks, A. R., Bradman, A., Harley, K., Barr, D. B., Johnson, C., Morga, N., & Jewell, N. P. (2007). Organophosphate pesticide exposure and neurodevelopment in young Mexican-American children. *Environmental Health Perspectives*, 115(5), 792–798. https://doi.org/10.1289/ehp.9828
- European Commission. (2012). *Toxicity and assessment of chemical mixtures*. European Union: Scientific Committee on Health and Environmental Risks.
- Farmer, C., Golden, C., & Thurm, A. (2016). Concurrent validity of the differential ability scales, second edition with the Mullen Scales of Early Learning in young children with and without neurodevelopmental disorders. *Child Neuropsychology*, 22(5), 556– 569. https://doi.org/10.1080/09297049.2015.1020775
- Fawzi, W. W., Rifas-Shiman, S. L., Rich-Edwards, J. W., Willett, W. C., & Gillman, M. W. (2004). Calibration of a semiquantitative food frequency questionnaire in early pregnancy. *Annals of Epidemiology*, 14(10), 754–762. https://doi.org/10.1016/j. annepidem.2004.03.001
- Fourth National Report on Human Exposure to Environmental Chemicals. (2009). Center for Disease Control and Prevention.
- Furlong, M. A., Barr, D. B., Wolff, M. S., & Engel, S. M. (2017). Prenatal exposure to pyrethroid pesticides and childhood behavior and executive functioning. *Neurotoxicology*, 62, 231–238. https:// doi.org/10.1016/j.neuro.2017.08.005
- Furlong, M. A., Engel, S. M., Barr, D. B., & Wolff, M. S. (2014). Prenatal exposure to organophosphate pesticides and reciprocal social behavior in childhood. *Environment International*, 70, 125–131. https://doi.org/10.1016/j.envint.2014.05.011
- Garay, P. A., & McAllister, A. K. (2010). Novel roles for immune molecules in neural development: Implications for neurodevelopmental disorders. *Frontiers in Synaptic Neuroscience*, 2, 136. https://doi. org/10.3389/fnsyn.2010.00136
- Gaspar, F. W., Harley, K. G., Kogut, K., Chevrier, J., Mora, A. M., Sjödin, A., & Eskenazi, B. (2015). Prenatal DDT and DDE exposure and child IQ in the CHAMACOS cohort. *Environment International*, 85, 206–212. https://doi.org/10.1016/j.envint.2015. 09.004
- Getz, K. D., Anderka, M. T., Werler, M. M., & Jick, S. S. (2016). Maternal pre-pregnancy body mass index and autism spectrum disorder among offspring: A population-based case-control study. *Paediatric and Perinatal Epidemiology*, 30(5), 479–487. https://doi. org/10.1111/ppe.12306
- Goines, P., & Van de Water, J. (2010). The immune system's role in the biology of autism. *Current Opinion in Neurology*, 23(2), 111–117. https://doi.org/10.1097/WCO.0b013e3283373514

- Hagmar, L., Wallin, E., Vessby, B., Jönsson, B. A., Bergman, A., & Rylander, L. (2006). Intra-individual variations and time trends 1991-2001 in human serum levels of PCB, DDE and hexachlorobenzene. *Chemosphere*, 64(9), 1507–1513. https://doi. org/10.1016/j.chemosphere.2005.12.054
- Hallmayer, J., Cleveland, S., Torres, A., Phillips, J., Cohen, B., Torigoe, T., Miller, J., Fedele, A., Collins, J., Smith, K., Lotspeich, L., Croen, L. A., Ozonoff, S., Lajonchere, C., Grether, J. K., & Risch, N. (2011). Genetic heritability and shared environmental factors among twin pairs with autism. *Archives of General Psychiatry*, 68(11), 1095–1102. https://doi.org/10.1001/ archgenpsychiatry.2011.76
- Hamra, G. B., Lyall, K., Windham, G. C., Calafat, A. M., Sjödin, A., Volk, H., & Croen, L. A. (2019). Prenatal exposure to endocrinedisrupting chemicals in relation to autism spectrum disorder and intellectual disability. *Epidemiology*, 30(3), 418–426. https://doi. org/10.1097/ede.00000000000983
- Hu, Y., Chiu, Y. H., Hauser, R., Chavarro, J., & Sun, Q. (2016). Overall and class-specific scores of pesticide residues from fruits and vegetables as a tool to rank intake of pesticide residues in United States: A validation study. *Environment International*, 92-93, 294–300. https://doi.org/10.1016/j.envint.2016.04.028
- Jayaraj, R., Megha, P., & Sreedev, P. (2016). Organochlorine pesticides, their toxic effects on living organisms and their fate in the environment. *Interdisciplinary Toxicology*, 9(3–4), 90–100. https://doi.org/ 10.1515/intox-2016-0012
- Julvez, J., Méndez, M., Fernandez-Barres, S., Romaguera, D., Vioque, J., Llop, S., Ibarluzea, J., Guxens, M., Avella-Garcia, C., Tardón, A., Riaño, I., Andiarena, A., Robinson, O., Arija, V., Esnaola, M., Ballester, F., & Sunyer, J. (2016). Maternal consumption of seafood in pregnancy and child neuropsychological development: A longitudinal study based on a population with high consumption levels. *American Journal of Epidemiology*, 183(3), 169–182. https://doi.org/10.1093/aje/kwv195
- Korrick, S. A., & Sagiv, S. K. (2008). Polychlorinated biphenyls, organochlorine pesticides and neurodevelopment. *Current Opinion in Pediatrics*, 20(2), 198–204. https://doi.org/10.1097/MOP. 0b013e3282f6a4e9
- Kortenkamp, A. (2007). Ten years of mixing cocktails: A review of combination effects of endocrine-disrupting chemicals. *Environmental Health Perspectives*, 115(Suppl 1), 98–105. https://doi.org/ 10.1289/ehp.9357
- Laetz, C. A., Baldwin, D. H., Collier, T. K., Hebert, V., Stark, J. D., & Scholz, N. L. (2009). The synergistic toxicity of pesticide mixtures: Implications for risk assessment and the conservation of endangered Pacific salmon. *Environmental Health Perspectives*, 117(3), 348–353. https://doi.org/10.1289/ehp.0800096
- Li, C., Cheng, Y., Tang, Q., Lin, S., Li, Y., Hu, X., Nian, J., Gu, H., Lu, Y., Tang, H., Dai, S., Zhang, H., Jin, C., Zhang, H., Jin, Y., & Jin, Y. (2014). The association between prenatal exposure to organochlorine pesticides and thyroid hormone levels in newborns in Yancheng, China. *Environmental Research*, 129, 47– 51. https://doi.org/10.1016/j.envres.2013.12.009
- Li, Y.-F. (n.d.). Organochlorine pesticide emission inventory review. GEIA: Global Emissions InitiAtive. http://www.geiacenter.org/ access/geia-originals
- Lyall, K., Anderson, M., Kharrazi, M., & Windham, G. C. (2017). Neonatal thyroid hormone levels in association with autism spectrum disorder. *Autism Research*, 10(4), 585–592. https://doi.org/10. 1002/aur.1708
- Lyall, K., Croen, L., Daniels, J., Fallin, M. D., Ladd-Acosta, C., Lee, B. K., Park, B. Y., Snyder, N. W., Schendel, D., Volk, H., Windham, G. C., & Newschaffer, C. (2017). The changing epidemiology of autism spectrum disorders. *Annual Review of Public Health*, 38, 81–102. https://doi.org/10.1146/annurev-publhealth-031816-044318
- Lyall, K., Croen, L. A., Sjödin, A., Yoshida, C. K., Zerbo, O., Kharrazi, M., & Windham, G. C. (2017). Polychlorinated

biphenyl and organochlorine pesticide concentrations in maternal mid-pregnancy serum samples: Association with autism spectrum disorder and intellectual disability. *Environmental Health Perspectives*, *125*(3), 474–480. https://doi.org/10.1289/ehp277

- Lyall, K., Munger, K. L., O'Reilly, É. J., Santangelo, S. L., & Ascherio, A. (2013). Maternal dietary fat intake in association with autism spectrum disorders. *American Journal of Epidemiol*ogy, 178(2), 209–220. https://doi.org/10.1093/aje/kws433
- Maski, K. P., Jeste, S. S., & Spence, S. J. (2011). Common neurological co-morbidities in autism spectrum disorders. *Current Opinion in Pediatrics*, 23(6), 609–615. https://doi.org/10.1097/MOP. 0b013e32834c9282
- McCullough, M. L., Feskanich, D., Stampfer, M. J., Giovannucci, E. L., Rimm, E. B., Hu, F. B., Spiegelman, D., Hunter, D. J., Colditz, G. A., & Willett, W. C. (2002). Diet quality and major chronic disease risk in men and women: Moving toward improved dietary guidance. *The American Journal of Clinical Nutrition*, 76(6), 1261–1271. https://doi.org/10.1093/ ajcn/76.6.1261
- Mense, S. M., Sengupta, A., Lan, C., Zhou, M., Bentsman, G., Volsky, D. J., Whyatt, R. M., Perera, F. P., & Zhang, L. (2006). The common insecticides cyfluthrin and chlorpyrifos alter the expression of a subset of genes with diverse functions in primary human astrocytes. *Toxicological Sciences*, 93(1), 125–135. https:// doi.org/10.1093/toxsci/kfl046
- Mie, A., Andersen, H. R., Gunnarsson, S., Kahl, J., Kesse-Guyot, E., Rembiałkowska, E., Quaglio, G., & Grandjean, P. (2017). Human health implications of organic food and organic agriculture: A comprehensive review. *Environmental Health*, 16(1), 111. https:// doi.org/10.1186/s12940-017-0315-4
- Millenson, M. E., Braun, J. M., Calafat, A. M., Barr, D. B., Huang, Y. T., Chen, A., Lanphear, B. P., & Yolton, K. (2017). Urinary organophosphate insecticide metabolite concentrations during pregnancy and children's interpersonal, communication, repetitive, and stereotypic behaviors at 8 years of age: The home study. *Environmental Research*, 157, 9–16. https://doi.org/10.1016/ j.envres.2017.05.008
- Miller, M., Sun, S., Iosif, A. M., Young, G. S., Belding, A., Tubbs, A., & Ozonoff, S. (2021). Repetitive behavior with objects in infants developing autism predicts diagnosis and later social behavior as early as 9 months. *Journal of Abnormal Psychology*, *130*(6), 665–675. https://doi.org/10.1037/abn0000692
- Morgan, M. K. (2012). Children's exposures to pyrethroid insecticides at home: A review of data collected in published exposure measurement studies conducted in the United States. *International Journal of Environmental Research and Public Health*, 9(8), 2964– 2985. https://doi.org/10.3390/ijerph9082964
- Moser, V. C., Simmons, J. E., & Gennings, C. (2006). Neurotoxicological interactions of a five-pesticide mixture in preweanling rats. *Toxicological Sciences*, 92(1), 235–245. https://doi.org/10. 1093/toxsci/kfj189
- Mullen, E. M. (1995). *The Mullen scales of early learning*. American Guidance Service Inc.
- Newschaffer, C. J., Croen, L. A., Fallin, M. D., Hertz-Picciotto, I., Nguyen, D. V., Lee, N. L., Berry, C. A., Farzadegan, H., Hess, H. N., Landa, R. J., Levy, S. E., Massolo, M. L., Meyerer, S. C., Mohammed, S. M., Oliver, M. C., Ozonoff, S., Pandey, J., Schroeder, A., & Shedd-Wise, K. M. (2012). Infant siblings and the investigation of autism risk factors. *Journal of Neurodevelopmental Disorders*, 4(1), 7. https://doi.org/10.1186/1866-1955-4-7
- Pan, I. J., Daniels, J. L., Goldman, B. D., Herring, A. H., Siega-Riz, A. M., & Rogan, W. J. (2009). Lactational exposure to polychlorinated biphenyls, dichlorodiphenyltrichloroethane, and dichlorodiphenyldichloroethylene and infant neurodevelopment: An analysis of the pregnancy, infection, and nutrition babies study. *Environmental Health Perspectives*, 117(3), 488–494. https:// doi.org/10.1289/ehp.0800063

- Pesticide Data Program Annual Summary, Calendar Year 2013. (2014). *Pesticide data program annual summary*. U.S. Department of Agriculture's (USDA) Agricultural Marketing Service (AMS).
- Philippat, C., Barkoski, J., Tancredi, D. J., Elms, B., Barr, D. B., Ozonoff, S., Bennett, D. H., & Hertz-Picciotto, I. (2018). Prenatal exposure to organophosphate pesticides and risk of autism spectrum disorders and other non-typical development at 3 years in a high-risk cohort. *International Journal of Hygiene and Environmental Health*, 221(3), 548–555. https://doi.org/10.1016/j.ijheh.2018. 02.004
- Rauh, V. A., Perera, F. P., Horton, M. K., Whyatt, R. M., Bansal, R., Hao, X., Liu, J., Barr, D. B., Slotkin, T. A., & Peterson, B. S. (2012). Brain anomalies in children exposed prenatally to a common organophosphate pesticide. *Proceedings of the National Academy of Sciences of the United States of America*, 109(20), 7871– 7876. https://doi.org/10.1073/pnas.1203396109
- Raz, R., Kioumourtzoglou, M. A., & Weisskopf, M. G. (2018). Livebirth bias and observed associations between air pollution and autism. *American Journal of Epidemiology*, 187(11), 2292–2296. https://doi.org/10.1093/aje/kwy172
- Rider, C. V., Furr, J. R., Wilson, V. S., & Gray, L. E., Jr. (2010). Cumulative effects of in utero administration of mixtures of reproductive toxicants that disrupt common target tissues via diverse mechanisms of toxicity. *International Journal of Andrology*, 33(2), 443–462. https://doi.org/10.1111/j.1365-2605.2009.01049.x
- Rimm, E. B., Giovannucci, E. L., Stampfer, M. J., Colditz, G. A., Litin, L. B., & Willett, W. C. (1992). Reproducibility and validity of an expanded self-administered semiquantitative food frequency questionnaire among male health professionals. *American Journal* of Epidemiology, 135(10), 1114–1126; discussion 1127-1136. https://doi.org/10.1093/oxfordjournals.aje.a116211
- Risch, N., Hoffmann, T. J., Anderson, M., Croen, L. A., Grether, J. K., & Windham, G. C. (2014). Familial recurrence of autism spectrum disorder: Evaluating genetic and environmental contributions. *The American Journal of Psychiatry*, 171(11), 1206– 1213. https://doi.org/10.1176/appi.ajp.2014.13101359
- Roberts, E. M., English, P. B., Grether, J. K., Windham, G. C., Somberg, L., & Wolff, C. (2007). Maternal residence near agricultural pesticide applications and autism spectrum disorders among children in the California Central Valley. *Environmental Health Perspectives*, 115(10), 1482–1489. https://doi.org/10.1289/ehp. 10168
- Rosas, L. G., & Eskenazi, B. (2008). Pesticides and child neurodevelopment. *Current Opinion in Pediatrics*, 20(2), 191–197. https://doi.org/10.1097/MOP.0b013e3282f60a7d
- Sacrey, L. R., Zwaigenbaum, L., Bryson, S., Brian, J., Smith, I. M., Roberts, W., Szatmari, P., Vaillancourt, T., Roncadin, C., & Garon, N. (2021). Screening for behavioral signs of autism spectrum disorder in 9-month-old infant siblings. *Journal of Autism* and Developmental Disorders, 51(3), 839–848. https://doi.org/10. 1007/s10803-020-04371-0
- Sagiv, S. K., Harris, M. H., Gunier, R. B., Kogut, K. R., Harley, K. G., Deardorff, J., Bradman, A., Holland, N., & Eskenazi, B. (2018). Prenatal organophosphate pesticide exposure and traits related to autism spectrum disorders in a population living in proximity to agriculture. *Environmental Health Perspectives*, 126(4), 047012. https://doi.org/10.1289/ehp2580
- Schmidt, R. J., Kogan, V., Shelton, J. F., Delwiche, L., Hansen, R. L., Ozonoff, S., Ma, C. C., McCanlies, E., Bennett, D. H., Hertz-Picciotto, I., Tancredi, D. J., & Volk, H. E. (2017). Combined prenatal pesticide exposure and folic acid intake in relation to autism spectrum disorder. *Environmental Health Perspectives*, 125(9), 097007. https://doi.org/10.1289/ehp604
- Schmidt, R. J., Tancredi, D. J., Ozonoff, S., Hansen, R. L., Hartiala, J., Allayee, H., Schmidt, L. C., Tassone, F., & Hertz-Picciotto, I. (2012). Maternal periconceptional folic acid intake and risk of autism spectrum disorders and developmental delay in the CHARGE (CHildhood Autism Risks from Genetics and

Environment) case-control study. *The American Journal of Clinical Nutrition*, 96(1), 80–89. https://doi.org/10.3945/ajcn.110.004416

- Shelton, J. F., Geraghty, E. M., Tancredi, D. J., Delwiche, L. D., Schmidt, R. J., Ritz, B., Hansen, R. L., & Hertz-Picciotto, I. (2014). Neurodevelopmental disorders and prenatal residential proximity to agricultural pesticides: The CHARGE study. *Environmental Health Perspectives*, *122*(10), 1103–1109. https://doi.org/ 10.1289/ehp.1307044
- Shelton, J. F., Tancredi, D. J., & Hertz-Picciotto, I. (2010). Independent and dependent contributions of advanced maternal and paternal ages to autism risk. *Autism Research*, 3(1), 30–39. https://doi.org/ 10.1002/aur.116
- Sioen, I., den Hond, E., Nelen, V., van de Mieroop, E., Croes, K., van Larebeke, N., Nawrot, T. S., & Schoeters, G. (2013). Prenatal exposure to environmental contaminants and behavioural problems at age 7-8years. *Environment International*, 59, 225–231. https://doi.org/10.1016/j.envint.2013.06.014
- Soldin, O. P., Lai, S., Lamm, S. H., & Mosee, S. (2003). Lack of a relation between human neonatal thyroxine and pediatric neurobehavioral disorders. *Thyroid*, 13(2), 193–198. https://doi.org/10. 1089/105072503321319503
- Starek-&wiechowicz, B., Budziszewska, B., & Starek, A. (2017). Hexachlorobenzene as a persistent organic pollutant: Toxicity and molecular mechanism of action. *Pharmacological Reports*, 69(6), 1232–1239. https://doi.org/10.1016/j.pharep.2017.06.013
- Subar, A. F., Thompson, F. E., Kipnis, V., Midthune, D., Hurwitz, P., McNutt, S., McIntosh, A., & Rosenfeld, S. (2001). Comparative validation of the Block, Willett, and National Cancer Institute food frequency questionnaires: The Eating at America's Table Study. *American Journal of Epidemiology*, 154(12), 1089– 1099. https://doi.org/10.1093/aje/154.12.1089
- Vogt, R., Bennett, D., Cassady, D., Frost, J., Ritz, B., & Hertz-Picciotto, I. (2012). Cancer and non-cancer health effects from food contaminant exposures for children and adults in California: A risk assessment. *Environmental Health*, 11, 83. https://doi.org/ 10.1186/1476-069x-11-83
- von Ehrenstein, O. S., Ling, C., Cui, X., Cockburn, M., Park, A. S., Yu, F., Wu, J., & Ritz, B. (2019). Prenatal and infant exposure to ambient pesticides and autism spectrum disorder in children: Population based case-control study. *BMJ*, 364, 1962. https://doi.org/ 10.1136/bmj.1962
- Wang, Y., Chen, L., Wang, C., Hum, Y., Gao, Y., Zhou, Y., Shi, R., Zhang, Y., Kamijima, M., Ueyama, J., & Tian, Y. (2017). Association between organophosphate pesticide exposure and thyroid hormones in pregnant women. *Epidemiology*, 28(Suppl 1), S35– s40. https://doi.org/10.1097/ede.000000000000721
- Wetherby, A. M., Guthrie, W., Hooker, J. L., Delehanty, A., Day, T. N., Woods, J., Pierce, K., Manwaring, S. S., Thurm, A., Ozonoff, S., Petkova, E., & Lord, C. (2021). The early screening for autism and communication disorders: Field-testing an autismspecific screening tool for children 12 to 36 months of age. *Autism*, 25(7), 2112–2123. https://doi.org/10.1177/13623613211012526
- Willett, W. (2013). Nutritional epidemiology (3rd ed.). Oxford University Press.
- Willett, W. C., Sampson, L., Stampfer, M. J., Rosner, B., Bain, C., Witschi, J., Hennekens, C. H., & Speizer, F. E. (1985). Reproducibility and validity of a semiquantitative food frequency

questionnaire. American Journal of Epidemiology, 122(1), 51–65. https://doi.org/10.1093/oxfordjournals.aje.a114086

- Wolff, M. S., Deych, E., Ojo, F., & Berkowitz, G. S. (2005). Predictors of organochlorines in New York City pregnant women, 1998-2001. Environmental Research, 97(2), 170–177. https://doi. org/10.1016/j.envres.2004.07.014
- Woodruff, T. J., Zota, A. R., & Schwartz, J. M. (2011). Environmental chemicals in pregnant women in the United States: NHANES 2003-2004. *Environmental Health Perspectives*, 119(6), 878–885. https://doi.org/10.1289/ehp.1002727
- Xue, J., Zartarian, V., Tornero-Velez, R., & Tulve, N. S. (2014). EPA's SHEDS-multimedia model: Children's cumulative pyrethroid exposure estimates and evaluation against NHANES biomarker data. *Environment International*, 73, 304–311. https://doi.org/10. 1016/j.envint.2014.08.008
- Yamazaki, K., Araki, A., Nakajima, S., Miyashita, C., Ikeno, T., Itoh, S., Minatoya, M., Kobayashi, S., Mizutani, F., Chisaki, Y., & Kishi, R. (2018). Association between prenatal exposure to organochlorine pesticides and the mental and psychomotor development of infants at ages 6 and 18 months: The Hokkaido Study on Environment and Children's Health. *Neurotoxicology*, 69, 201–208. https://doi.org/10.1016/j.neuro. 2017.11.011
- Yau, V. M., Lutsky, M., Yoshida, C. K., Lasley, B., Kharrazi, M., Windham, G., Gee, N., & Croen, L. A. (2015). Prenatal and neonatal thyroid stimulating hormone levels and autism spectrum disorders. *Journal of Autism and Developmental Disorders*, 45(3), 719–730. https://doi.org/10.1007/s10803-014-2227-2
- Zwaigenbaum, L., Bauman, M. L., Fein, D., Pierce, K., Buie, T., Davis, P. A., Newschaffer, C., Robins, D. L., Wetherby, A., Choueiri, R., Kasari, C., Stone, W. L., Yirmiya, N., Estes, A., Hansen, R. L., McPartland, J., Natowicz, M. R., Carter, A., Granpeesheh, D., ... Wagner, S. (2015). Early screening of autism spectrum disorder: Recommendations for practice and research. *Pediatrics*, 136(Suppl 1), S41–S59. https://doi.org/10.1542/peds. 2014-3667D
- Zwaigenbaum, L., Brian, J. A., & Ip, A. (2019). Early detection for autism spectrum disorder in young children. *Paediatrics & Child Health*, 24(7), 424–443. https://doi.org/10.1093/pch/pxz119

## SUPPORTING INFORMATION

Additional supporting information may be found in the online version of the article at the publisher's website.

How to cite this article: Joyce, E. E., Chavarro, J. E., Rando, J., Song, A. Y., Croen, L. A., Fallin, M. D., Hertz-Picciotto, I., Schmidt, R. J., Volk, H., Newschaffer, C. J., & Lyall, K. (2022). Prenatal exposure to pesticide residues in the diet in association with child autism-related traits: Results from the EARLI study. *Autism Research*, *15*(5), 957–970. https://doi.org/10.1002/aur.2698